Author's response to reviews

Title: Rate of and Factors Associated with breakfast among Inner Mongolia Medical Students in China

Authors:

juan sun (cnsunjuan@yahoo.com.cn)
he yi (snj_lyh@126.com)
zhiyue liu (liuzhiyue@163.net)
jiang bian (bianjiang0476@163.com)
Yuki Eshita (yeshita@oita-u.ac.jp)
gaiyue li (ligaimei1966@163.com)
qing zhang (cfzhangqing@yahoo.com.cn)

Version: 2 Date: 30 December 2012

Author's response to reviews: see over
Dear Dr Kawika Liu,

We thank you very much for giving us an opportunity to revise our manuscript and we appreciate the comments and suggestions reviewers have made regarding our manuscript.

We have made revisions, which are highlighted in yellow. Attached please find the revised manuscript, our questionnaire, and response to reviewers’ comments.

We look forward to hearing from you.

Yours sincerely,

Juan Sun